Author's response to reviews

Title: Posterior Leukoencephalopathy following Repair of an Ileo-cecal Anastomosis Breakdown: A Case Report and Review of the Literature

Authors:

Pascal O Zinn (pzinn@bidmc.harvard.edu)
Rivka R Colen (RRCOLEN@partners.org)
Ekkehard M Kasper (ekasper@bidmc.harvard.edu)
Clark C Chen (ccchen1@partners.org)

Version: 3 Date: 4 November 2010

Author's response to reviews:

All the requested changes have been addressed:

Major revisions (we require the author to make these changes)

Email addresses: We note that the email address of Clark C Chen is different on the submission system to the manuscript. Please ensure email addresses are identical between the submission system and the manuscript.

Patient ID: Please ensure the patient identification (age, gender and ethnicity) is included in the abstract in the manuscript and the abstract on the submission system.

Authors' contributions: Please indicate that all authors read and approved the final manuscript.

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Affiliations: Please remember to include the country in the affiliation details

Structure: Please ensure the Competing interests section precedes the Consent section

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.